Author’s response to reviews

Title: Furuncular Myiasis Of The Breast Caused by the African Tumbu Fly (Cordylobia anthropophaga). Adisa Adeyinka Charles* Mbanaso A.U** *Surgery Department Abia State University Teaching Hospital Aba, Abia state. E mail:adisayinka@yahoo.com **Surgery Department Abia State University Teaching Hospital Aba, Abia state. E mail:ausymbanaso@yahoo.com

Authors:

Dr Adeyinka C Adisa dr (adisayinka@yahoo.com)
Augustine U Mbanaso dr (ausymbanaso@yahoo.com)

Version: 2 Date: 7 Nov 2003

Dear Editor,
I have just made a revision to the title page of the article by adding the names, contact addresses and the e-mail addresses of the authors.
Sincerely,
Dr Adisa.